Format for ANSWERING REVIEWERS

May 18, 2013

Dear Editor,

Please find enclosed the edited manuscript in Word format (3469-edited).

Title: Transcatheter arterial chemoembolization followed by immediate radiofrequency ablation for solitary large hepatocellular carcinoma

Author: Zhi-Jun Wang, Mao-Qiang Wang, Feng Duan, Peng song, Feng-Yong Liu, Zhong-Fei Chang, Yan Wang, Jie-Yu Yan, Kai Li,

Name of Journal: World Journal of Gastroenterology

ESPS Manuscript NO: 3469

The manuscript has been improved according to the suggestions of reviewers: 1 Format has been updated

- 2 Revision has been made according to the suggestions of the reviewer
 - (1) Articles of reference have been updated and supplemented
 - (2) Figures have been updated and adjusted.

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the World Journal of Gastroenterology.

Sincerely yours,

Wang Zhijun, MD, PhD
Department of interventional radiology
PLA General Hospital
28 Fuxing Road ,Hai dian District ,Beijing,China,100853

Fax: +86-010-6693-6327

E-mail: wangzj301hospital@163.com

